Author's response to reviews

Title: Testing for sexually transmitted infections in general practice: A cross-sectional study

Authors:

Katharine E Sadler (katsaddle@hotmail.com)
Nicola Low (low@ispm.unibe.ch)
Catherine H Mercer (cmercergum.ucl.ac.uk)
Lorna Sutcliffe (l.s.sutcliffe@qmul.ac.uk)
M Amir Islam (m.a.a.islam@www.ac.uk)
Shuja Shafi (msshafi1@googlemail.com)
M Gary Brook (Gary.Brook@nwh.nhs.uk)
Helen Maguire (helan.maguire@hpa.org.uk)
Patrick J Horner (Paddy.Horner@bristol.ac.uk)
Jackie A Cassell (j.cassell@bsms.ac.uk)

Version: 3 Date: 20 October 2010

Author's response to reviews:

Dear Editors,

We are submitting a revised version with responses to the Editors' requests, as follows:

1. List of authors has been changed, as requested;
2. Conclusion has been added after Discussion;
3. Tables re-numbered as requested, and citations in manuscript changed;
4. Background section correctly labelled;
5. Figures prepared in accordance with guidelines;
6. Typography checked. Please can you check the reference list format? I have pasted in the correctly formatted references, but Reference Manager tries to re-format. Apologies for this.

I hope that our submission now meets all requirements.

Kind regards,

Nicola Low